Author's response to reviews

Title: Mobile phone use and stress, sleep disturbances, and symptoms of depression among young adults - a prospective cohort study

Authors:

Sara Thomée (sara.thomee@amm.gu.se)
Annika Härenstam (annika.harenstam@av.gu.se)
Mats Hagberg (mats.hagberg@amm.gu.se)

Version: 6 Date: 19 January 2011

Author’s response to reviews:

Dear Editors of BMC Public Health,

Table 1 has been changed in accordance with the requested revision. We have also made a minor typographical change in Figure 1 (Calls and SMS in italics).

We hope the changes are satisfactory and that the manuscript is ready for publication.

Yours sincerely,

Sara Thomée (on behalf of the authors)

Occupational and Environmental Medicine
Department of Public Health and Community Medicine
University of Gothenburg
Gothenburg
Sweden